

Gender-related factors affecting health seeking for neglected tropical diseases: findings from a qualitative study in Ethiopia

Alexandra Wharton-Smith 1*, Christian Rassi¹, Esey Batisso², Giuseppina Ortu 1*, Rebecca King³, Misganu Endriyas⁴, Helen Counihan¹, Prudence Hamade¹, Dawit Getachew⁵

- 1 Malaria Consortium, London, United Kingdom, 2 Malaria Consortium, Hawassa, Ethiopia, 3 The Nuffield Centre for International Health & Development, University of Leeds, Leeds, United Kingdom, 4 Regional Health Bureau, Hawassa, Ethiopia, 5 Malaria Consortium, Addis Ababa, Ethiopia
- * alexandra.wharton-smith@lshtm.ac.uk



OPEN ACCESS

Citation: Wharton-Smith A, Rassi C, Batisso E, Ortu G, King R, Endriyas M, et al. (2019) Gender-related factors affecting health seeking for neglected tropical diseases: findings from a qualitative study in Ethiopia. PLoS Negl Trop Dis 13 (12): e0007840. https://doi.org/10.1371/journal.pntd.0007840

Editor: Sally Theobald, Liverpool School of Tropical Medicine, UNITED KINGDOM

Received: June 6, 2019

Accepted: October 11, 2019

Published: December 12, 2019

Copyright: © 2019 Wharton-Smith et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the manuscript and its Supporting Information files.

Funding: The study was co-funded by COMDIS-HSD (comdis-hsd.leeds.ac.uk), a research consortium funded by UK aid from the UK government (www.gov.uk/government/organisations/department-for-international-development). The views expressed in this manuscript do not necessarily reflect the funder's

Background

Abstract

Despite known gender-specific differences in terms of prevalence, transmission and exposure to neglected tropical diseases (NTDs), there is limited discussion of the influence of gender in NTD programmes and interventions. There is a paucity of research on how gender interacts with NTD service provision and uptake. This study, part of broader implementation research in Ethiopia, applied a gender lens to health seeking for five NTDs: lymphatic filariasis, podoconiosis, schistosomiasis, soil-transmitted helminth infection and trachoma.

Methodology/principal findings

The study was conducted in a district of the Southern Nations, Nationalities, and Peoples' Region of Ethiopia where the five NTDs are prevalent. A qualitative methodology was adopted to explore participants' perspectives and experiences. Data generation methods included 20 interviews and four focus group discussions. Community members, volunteer Health Development Army leaders, Health Extension Workers and a range of health workers at the health post, health centre and hospital level (n = 59) were purposively sampled. Interviews and focus group discussions were audio recorded, transcribed verbatim into English then analysed through open coding, drawing on constant comparative methods.

Gender related factors affected care seeking for NTDs and were described as reasons for not seeking care, delayed care seeking and treating NTDs with natural remedies. Women faced additional challenges in seeking health care due to gender inequalities and power dynamics in their domestic partnerships. Participants recommended raising community awareness about NTDs, however this remains problematic due to gender and social norms around appropriate discourse with members of the opposite gender.



official positions and policies. The funder had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Conclusions/significance

The findings from this study provide crucial insights into how gender interacts with accessing health services, at different levels of the health system. If we are committed to leaving no one behind and achieving universal health coverage, it is essential to address gender disparities to access and utilisation of interventions delivered by national NTD programmes.

Author summary

This study provides insights into how gender influences peoples' access to health services for neglected tropical diseases (NTDs). There is considerable evidence to show that women and girls suffer a greater NTD burden due to their higher levels of poverty, lower education and social status. The psychological and social consequences of NTDs also tend to be worse for women. There is an urgent need to integrate NTDs into basic healthcare services, yet there is little evidence that can support a gender-based approach of how best to do so. This study addresses that gap by exploring how gender affects health seeking behaviour for NTDs in Ethiopia through a series of focus group discussions and interviews with community members and health service providers. The findings provide new insights into why women might not seek care or delay care seeking. In particular, there are reservations about disclosing diseases that affect "hidden" body parts, for example in the case of schistosomiasis, a parasitic infection that can affect the urogenital system. Men, on the other hand, may delay seeking care if they are affected by hydrocele, scrotal swelling that can be caused by lymphatic filariasis. There is therefore a need to adopt a gendered approach to NTD health service delivery.

Introduction

There is limited discussion of the role of gender in neglected tropical disease (NTD) programmes, despite known gender-specific differences in terms of prevalence [1-3], transmission and exposure to NTDs [4]. These differences have been attributed to biological vulnerability and gender roles [5–8]. Often, women and girls experience a greater share of the NTD burden due to their disproportionate poverty, illiteracy, lower education and social status [9,10]. For example, in societies where women are the primary caregivers, they experience an increased risk of exposure and severe disease. A well-documented example is trachoma, where female caregivers come into more frequent contact with infected children than male caregivers and are thus more likely to become infected themselves [5,11]. In the case of podoconiosis, women have been found to be more at risk of acquiring the disease because they are less likely to own or wear shoes, reflecting cultural traditions and because available shoes were perceived to be more suitable for men [12,13]. Other studies have remarked on the underestimation and underreporting of NTDs in women, specifically schistosomiasis and lymphatic filariasis (LF), as diagnosis of these diseases can require women to comply with activities that are perceived as culturally inappropriate or taboo, such as providing a urine or stool sample or allowing intimate physical examination [14,15].

The psychological and social consequences of NTDs, which can be mutually reinforcing, especially of diseases that cause disfigurement, also tend to be worse for women [16]. Several studies of different NTDs found that males reported significantly lower levels of enacted and

experienced stigma [17–19]. Other studies have illustrated how women's experiences of physical impairment from NTDs exacerbate existing socio-cultural and economic inequalities [20–23]. Physical disfigurement is particularly detrimental for women, as their social acceptance and marriageability may rely on their physical appearance more than men's [24,18,25], which can lead to social reclusiveness [25, 26]. Women are also at an increased risk of NTD-related mental distress and intimate partner violence (IPV) [27, 16, 28]. A recent study in Ethiopia found that podoconiosis in women increases the frequency and severity of IPV, including physical, psychological and financial violence and was attributed to "men's frustration at being married to an 'unhealthy' woman who cannot always fulfil household and work expectations" [28]. Men, on the other hand, are uniquely affected by stigma associated with the disability of hydrocele, which has been linked with dissatisfaction with patients' and partners' sex lives, and, additionally, a belief that it causes impotence [24].

While gender-related differences in terms of NTD burden are well documented, there is a paucity of research on how gender interacts with NTD service provision and uptake. Of the limited literature available, studies have mainly explored access to mass drug administration (MDA) and how gender relates to the experience of being an NTD volunteer [29–36]. For example, that female NTD volunteers uniquely contend with the challenges of balancing work and family commitments [31]. In terms of MDA coverage, a study in Uganda found that men may miss out on treatment as they were more likely to be away from their homes due to occupational roles, whilst some Community Drug Distributors (CDDs) withheld drugs from pregnant or breastfeeding women due to drug safety concerns [29]. However, a recent study of MDA coverage in 16 countries suggested that NTD programmes "consistently achieve at least equal levels of coverage for women" [30]. Study findings on NTD volunteer performance, specifically, the effectiveness and coverage of MDA by female versus male NTD volunteers and recipient compliance, varied by setting and treatment [31–37].

Few studies have addressed gender and health care seeking for NTDs. This literature highlights gender inequalities in terms of delayed care seeking [16, 5], and coverage of treatment and services [18, 19, 38, 39]. The majority of studies have concluded that female gender is disadvantageous [40-42]. However, much of the research on gender and NTDs to date has neglected the experience of men [43,44].

There is a clear research need for in-depth, qualitative exploration of gender as a factor in health seeking for NTDs to inform health programme design and improve gender equity, especially in the context of the "leaving no one behind" agenda [45] and achieving universal health coverage (UHC) [46]. As we move beyond the 2020 NTD roadmap targets [47], there is a need to integrate NTDs into routine primary health care (PHC) services, yet little evidence exists on how this can be designed taking gender-specific health seeking behaviours into account.

This qualitative study aimed to address this gap by applying a gender lens to health seeking for five NTDs: lymphatic filariasis, podoconiosis, schistosomiasis (intestinal and urogenital), soil-transmitted helminth (STH) infections and trachoma. The data presented in this paper was collected as part of a broader implementation research study. The findings provide crucial insights on how gender interacts with accessing care for NTDs at different levels of the PHC system in Ethiopia.

Methods

Study design

In 2017 and 2018, we conducted the research in collaboration with the Ethiopian Federal Ministry of Health (FMoH) and the Regional Health Bureau (RHB) in the Southern Nations, Nationalities, and Peoples' Region (SNNPR). The broader study explored feasibility and

acceptability of an intervention designed to strengthen detection, management and recording of the five target NTDs at the different levels of the PHC system in Ethiopia: hospital, health centre, health posts and the community volunteer Health Development Army (HDA). The intervention comprised the following components (see S2 Appendix for a more detailed description of the intervention, informed by the template for intervention description and replication (TIDieR) checklist and guide) [48]:

- Harmonised case definitions for the five target NTDs
- Definition of roles and responsibilities for the detection, management and recording of target NTDs at the different levels of the PHC system
- Training for health workers and supervisors on case definitions, roles and responsibilities and provision of practical job aids
- Training for HDA volunteers and provision of simple visual tools to detect community members with signs and symptoms of NTDs and encourage them to seek care

As part of a systematic evaluation of the intervention, health workers, HDA volunteers and community members were invited to participate in focus group discussions (FGDs) and key informant interviews (KIIs). The objectives of this study were to capture participants' perceptions of the intervention, as well as their experience of NTDs and NTD care more generally. Broader evaluation findings have been published elsewhere [49]; this paper focuses on data relating to gender and health care seeking. The reporting of qualitative methods in this paper has been cross-checked with the consolidated criteria for reporting qualitative research (COREQ) [50].

Study setting

The study was conducted in a rural *kebele* (the smallest administrative structure within a district) located in SNNPR, the third largest administrative region of Ethiopia and the country's most diverse in terms of language and ethnicity. Most of the population reside in rural areas, where the main economic activity is agriculture. The language most widely spoken in the study area is Sidamigna, the language spoken by the Sidama people. All five target NTDs are endemic in the study area, which was selected in consultation with RHB and for operational reasons. Health centres were purposively sampled based on whether they were active at the time of data collection and are assumed to be representative of similar health centres in rural areas in Ethiopia. All of the Health Extension Workers (HEW) and HDA leaders included in the study area were female, in line with national policy. The following inclusion criteria were applied to determine health centres' eligibility for inclusion in the study:

- Access to electricity (grid, solar or generator)
- Access to clean water and acceptable level of sanitation (e.g. presence of latrines and wells)
- Availability of rooms to perform simple laboratory analyses
- Reported cases of target NTDs in the recent past
- Not enrolled in other research studies
- Does not receive dedicated support from non-government organisations (NGO)
- Does not expect disruption to service delivery during intervention period

Of the six health centres in the study area, three were excluded based on the research team's knowledge of their location and circumstances. The remaining three health centres were visited by the study in June 2017 to determine their eligibility. Only one health centre in the study area met all inclusion criteria and was therefore selected as the study site. It serves a population of approximately 21,000. The health centre is overseen by a General Hospital, which was also included in the study. There was one active and functional health post in the health centre's catchment area at the time of data collection, which was staffed by Health Extension Workers (HEWs), a cadre of salaried, female community health workers. The health post supports a network of community-based female HDA volunteers.

All health workers at the health centre and hospital levels responsible for the detection, management and recording of NTDs in the study area were invited to participate in the study, along with all HEWs and HDA volunteers in the catchment area.

Data collection

Study tools for FGDs and KIIs were developed and tailored to each strata of the sample (see <u>\$3</u> Appendix). Semi-structured topic guides were used to elicit data on the following themes:

- NTD management at the community, health post, health centre and hospital level
- Gender
- Stigma
- Perceptions of the different components of the intervention

All topic guides were developed in English by CR and EB, and subsequently translated into Amharic (the country's official language) by a professional translator. During FGDs with HDA volunteers and community members, researchers translated verbally into Sidamigna, the local language commonly spoken in the study area. The topic guides could not be pre-tested. However, the transcripts were reviewed by CR at regular intervals and the topic guides subsequently refined based on emerging themes and gaps in the data.

Data collection was conducted between August and September 2018 by six field researchers (two female, four male) who were recruited for this exercise from local and regional health authorities, health teaching institutes and health consultancy firms. All except one had a master's degree in public health or related field. All researchers had previously been involved in field research and, again with one exception, had some prior experience of conducting qualitative research. All researchers attended a three-day training conducted by CR, which introduced the study design and objectives, an overview of qualitative research principles and techniques, ethical considerations and opportunities to practice qualitative research skills. Field work procedures and topic guides were discussed in detail, and the accuracy of the translations into Amharic was checked. While in the field, the researchers worked in pairs: one conducted the KII or moderated the FGD, whilst the other assisted and recorded notes. The researchers had regular quality assurance meetings with CR during the fieldwork to refine interview techniques and discuss edits to the topic guides.

Sampling of participants

Key informant interviews. Semi-structured interviews were used to explore themes in more depth [51] with health workers and laboratory staff in a private setting. This method minimised service disruption at participating health facilities, as FGDs would require several health workers to be away from their work station simultaneously. Conducting KIIs also offered the benefit of focussing on health workers' unique duties, as these differed across

occupational groups. Health staff at the health centre and hospital levels were identified in consultation with the acting facility in-charge and approached on the day of data collection based on their availability. Only health workers with responsibility for detecting, managing and recording NTDs were eligible, with separate additional sampling criteria for each level of PHC system, across specialties and departments. At the health centre level, this included:

- At least one health worker from each of the following sections: Paediatrics, Outpatient Department (OPD), *kebele*/Health Post Support
- At least two nurses
- One Health Officer
- Two laboratory technicians
- One Health Management Information System (HMIS) focal person
 At the hospital level, the following were sampled:
- At least one representative from each of the following case teams/wards was sought: paediatric, under-five OPD, adult OPD, Triage and Medical
- Two representatives from each of the following cadres: Health Officer, nurse, doctor
- Two laboratory staff
- One HMIS focal person

At both levels, the team aimed to sample roughly equal numbers of female and male participants. Within those criteria, sampling was based on convenience. At health post level, only two

Table 1. Interview sample by primary healthcare system level, cadre and sex.

Primary healthcare system level	Cadre	Sex
Hospital	Laboratory technician	Female
Hospital	Laboratory technician	Male
Hospital	Medical doctor, Triage	Male
Hospital	Nurse, Adult Outpatient Department	Male
Hospital	Health Officer, Triage	Male
Hospital	Health Officer, Triage	Female
Hospital	Medical Doctor, Paediatrics	Male
Hospital	Nurse, Triage	Female
Hospital	HMIS Focal Person	Male
Health Centre	Nurse, Kebele Health Post Support	Male
Health Centre	Health Officer, Outpatient Department	Female
Health Centre	Health Officer, Emergency and Kebele support	Male
Health Centre	HMIS Focal Person	Male
Health Centre	Midwife, Kebele/Health Post Support, Family Planning and Women Care	Female
Health Centre	Nurse, Kebele/Health Post Support	Male
Health Centre	Nurse, Paediatrics	Female
Health Centre	Laboratory technician	Male
Health Centre	Laboratory technician	Male
Health Post	Health Extension Worker	Female
Health Post	Health Extension Worker	Female

https://doi.org/10.1371/journal.pntd.0007840.t001

HEWs were included in the study, and both were interviewed. In total, KIIs were conducted with 20 health workers and laboratory staff: 9 at the hospital level, 9 from health centres and 2 from the health post (Table 1).

Focus group discussions. FGDs were conducted with community members and HDA volunteers. This included two FGDs with community members (one male group with 10 participants and one female group with 9 participants), and two FGDs with HDA volunteers (all female, with 10 participants in each). This type of data collection facilitated interaction between participants and the expression of norms and consensus among peers [52]. Additionally, FGDs were considered appropriate for these strata given the large number of potential participants in the study area.

Separate FGDs were conducted with female and male community members to encourage open discussion. FGDs with community members comprised of a culturally appropriate mix of participants who were considered influential in terms of community members' health seeking behaviour. This included *kebele* leaders, religious leaders and representatives from women's groups. Only HDA volunteers known to be active were recruited for the FGDs with this target group. The sample included HDA members from different villages in each FGD. All FGD participants were identified in consultation with HEWs and health centre staff.

One deviation from the sampling criteria was the accidental inclusion of three female HDA volunteers in the FGD with female community members. In this paper, participants from that FGD whose quotes are reported are accurately tagged as either "HDA" or "Community Member".

Data analysis

All KIIs and FGDs were audio recorded, translated and transcribed verbatim into English by the researchers who had conducted the KII or FGD immediately after data collection.

CR developed an initial coding frame based on close reading of the transcripts. The coding frame was discussed and agreed with RK based on close reading of a sample of five transcripts with different target groups. All transcripts were analysed by AWS using a modified thematic content analysis approach, which involved inductive identification of themes through open coding by close line-by-line interrogation, drawing on constant comparative methods [53]. Emerging themes were discussed with CR who also reviewed and commented on an early draft report which was written by AWS and later reviewed by EB, HC and PH. Data saturation was likely reached on the themes covered in the topic guides. Data extracts used in the report are tagged with the data source (KII or FGD), participant type (e.g. community member), and sex.

Note that the authors recognise the distinction between "sex" as a biologically determined classification based on chromosomal derived reproductive organs, and "gender" as an individual's self-identity and self-representation, which translates into a social, cultural and legal status. However "sex", specifically described as female or male, is the standard distinction used in the majority of the scientific literature to date. This traditional classification was also used by the data collectors and the participants in this study when discussing gender, which is reflected in the way gender is discussed in this paper.

Ethics

Ethical approval for the study was granted by the Ethiopian Southern Nations Nationalities and Peoples Regional Health Bureau Research Ethical Review Committee (reference number PMB-19/22566) and the University of Leeds Faculty of Medicine and Health Research Office, School of Medicine Research Ethics Committee (reference number: MREC16-146). Informed



written consent was given by all participants, who were adults (aged 18 or above). The data was anonymised prior to analysis to protect confidentiality.

Results and discussion

Local conceptualisation and gendering of NTDs

Understanding local beliefs and discourse on disease can shed light on how NTDs are conceptualised, classified and experienced *in situ*. Discussions with both community members, HDA leaders and HEWs revealed a low awareness of "neglected tropical diseases" in the manner that they are named and conceptualised in biomedicine or classified in public health. Community members sometimes described and named diseases in the local language with symptoms that were similar to those of the five NTDs targeted by the intervention. Participants generally discussed and categorised diseases in terms of the symptoms they observed or experienced and the body parts that were affected. Conceptualising diseases by body part meant that those involving the genitals or breasts were treated as gender-specific.

"Daniichu Xibba [literally "the disease of the elephant"]: a disease that makes the legs swell for men and a disease that makes women have swollen breasts." G17, Male Community Member, FGD

"I heard that the disease that affects women's wombs makes the women infertile. The disease is transmitted by a snail that is found in the water. When people come into contact with the water containing the snail and the snail enters the body through people's skin, it causes men to urinate red urine and women to be infertile." G17, Male Community Member, FGD

This local sorting of diseases into categories based on symptoms and affected body parts by community members contrasts with how the World Health Organisation and public health community groups diseases together according to different criteria from a global health perspective. Therefore, it is important to understand community health beliefs and engage in dialogue as part of NTD interventions.

When discussing the five target NTDs, participants also implicitly sorted these diseases into those which affected the genitals or "hidden" body parts (urogenital schistosomiasis, LF) from those that did not (trachoma, soil-transmitted helminth infection, intestinal schistosomiasis and podoconiosis). "Hidden" body parts tended to refer to genitals in both genders and breasts in women; body parts that may engage in sexual activity, which induced a social taboo. This distinction is important as the diseases which were perceived as affecting "private areas" were negotiated differently in terms of interactions with health services compared to diseases affecting other non-private, "visible" body parts due to this sexual, social taboo.

Participants explained that it is more socially acceptable to speak about diseases which affect body parts that are "visible" compared to diseases that affect body parts that are "hidden" or that are "sex organ[s]." It is noted that for a body part to be considered "hidden" participants were not necessarily referring to body parts that would conventionally be covered by clothing. The term was also used euphemistically to body parts that may play a role in sexual contact. Similarly, for "visible," this was more used to mean socially acceptable to discuss openly, between genders, rather than a body part that is physically visible in public. For the diseases related to the genitals or urinary tract, a few participants suggested that sufferers may hide their disease or be afraid to discuss it even though the disease might be common in the community.

"If the disease is around the genital area, they are afraid of talking with men. And sometimes diseases around women's womb are not easily identified because women hide the disease fearing their secret may be revealed." P01, Female HEW, Interview

"You know, there are people who are affected by urinary diseases. A lot of people are affected by those diseases and some even had surgery. Even I am affected by this disease. You see, the disease very much exists in the community, but no one dares to speak frankly about it. People do not talk about it with others, especially when the disease is related to the genital area. They hide the disease from the people around them." G12, Male Community Member, FGD

HEWs and HDA volunteers responsible for raising awareness about these diseases in the community lamented the challenge of asking community members about their "hidden" body parts or urination. The reluctance to discuss hidden body parts hinders open dialogue on NTDs

G02: Those body parts are not seen. They are hidden. The question is, how can we ask them?

G03: This picture shows a man's swollen sex organ. And you men, you keep that swollen thing inside your trousers. So how can we ask him? I think this will be difficult.

Facilitator: Have you tried asking people by showing this picture [shows picture in the pictorial tool]?

G04: It will be very difficult to ask people who have a swelling in that area.

G05: If a woman has a swollen breast, we can ask her about her pain during our coffee ceremonies.

G06: For example, if people have pain in the breast area, people may not understand...For those who have breast pain or pain in the uterus...it is difficult to ask those who have swollen sex organs. For the others, we can ask them what kind of pain they have because the breast area is visible and also the abdomen is visible.

Female HDA leaders, FGD

HDA volunteers acknowledged the reluctance to speak about these diseases and emphasised the importance of public discussion to raise awareness and encourage disclosure.

"When we teach about the disease in public to the community, those who hide their disease can disclose easily and get treatment for their diseases. People may hide this kind of disease but understanding about the disease can help us to disclose and make people aware of the diseases." E07, Female HDA leader, FGD

Both female and male participants explained that talking about genitals, even in the context of disease that affect these body parts is socially taboo and culturally unacceptable between women and men, even if they are community health workers. This rule applied to both genders. It was not deemed socially acceptable for men to bring the topic up with women, nor for women to discuss such diseases or body parts with men. An exception to this was that it was deemed acceptable to discuss any body part or disease affecting urination when speaking



about children, in contrast to sexually mature adults, possibly because such body parts in adults are related to sexual contact.

Talking openly about diseases that affect hidden areas with members of the opposite sex, regardless of profession, challenged socio-cultural norms and for women, risked the social consequences of being perceived as "ill-mannered."

"[After some hesitation] I saw a man with a swollen scrotum [all participants laughed]. At that time, we raised the question to the Health Extension Workers saying, 'You know, it is difficult to disclose and discuss this kind of issue with a man.' You know, in our culture, women do not dare to talk with a man even about usual business. So how can we teach and discuss this kind of issue? . . . culturally, we find it difficult to reach the men with such a problem, because we fear to talk to men." E03, Female HDA leader, FGD

"You know, female Health Development Army leaders may find it challenging to teach and detect the diseases that affect the male sex organs. I am absolutely sure that they are even afraid of naming those diseases that the males have. For example, they will be afraid to name the disease of the scrotum because in our culture, if the female names this disease frankly, she will be considered an ill-mannered woman." G17, Female Community Member, FGD

The reluctance to speak openly about hidden body parts and the NTDs that affect them has implications for raising awareness and care seeking, especially as all HDA volunteers are female. Because of how NTDs were conceptualised, and in some cases, gendered, discussing some NTDs at the community level as envisaged by our intervention is challenging.

HDA leaders stressed the importance of educating women on diseases that can affect their sexual and reproductive organs and promoting open discussion, despite cultural taboos, to encourage timely uptake of health services.

"If women are educated, they teach those women who have problems with their uterus to get better treatment. And also by learning all of this, we prevent disease and death among children and adults from neglected diseases. We also encourage those women who hide their problems to talk about their problems." E18, Female HDA leader, FGD

Suggestions from participants on how to communicate information about NTDs affecting men included training male community volunteers; communicating health messages to men through their wives, although this may be problematic if the man suspects a sexually transmitted infection; or family members and through male networks in the community, and to involve men in training on NTDs. The acceptability and feasibility of these proposed interventions would need to be explored through further research.

"For people with swollen testicles, it would be better to train males to educate those men. I'm afraid that will not be practical for us. It would be better to educate some men together with us. It is important to identify people with this problem through them. Because they may hide their problems from us women—they may hide the swelling." E18, Female HDA leader, FGD

"But the Health Extension Worker replied to us that a person with such a problem, you can approach the man through a family member of the person. For example, if you discuss the issue with the wife of that individual, that woman can directly talk to him and he can disclose his issue." E03, Female HDA leader, FGD



Hiding the disfigurement caused by NTDs was not limited to diseases affecting the breasts and genitals due to sexual connotations, which induced a social taboo; female community members also shielded other affected body parts from public view, as the disfigurement was perceived as a separate, additional, cultural taboo. This suggests that any swollen body part, for women at least, may become a "hidden" area because of the stigma associated with NTDs, whether related to sex or disfigurement, and may hinder detection of cases and care seeking.

E17: There is one person who has a swollen scrotum in my village, but he always hides it.

E15: In my village people also have eye problems and there is an old woman with leg swelling. There's another lady with a swollen leg, but she hides it under long dresses, so people don't see it as it is seen as a taboo." Female HDA leaders, FGD

Gender and health care seeking for NTDs

Gender as it related to health care seeking was generally discussed along two themes: 1) shame and fear of disclosing certain NTD symptoms and 2) power dynamics.

Shame and fear of disclosing symptoms. People discussed a number of gender related factors as challenges to care seeking. In some cases, these were discussed as deterring people from seeking care at all; in other cases, they were seen as barriers to timely care seeking. Women found it more challenging to disclose swellings of the limbs due to shame and social taboos. This finding resonates with the literature on NTD-related disfigurement, stigma and lowered social status for women compared to men [42]. The stigma around hidden body parts poses challenges to HDA leaders' and HEWs' work in identifying and discussing symptoms that affect private body parts, particularly in discussions between female community health workers and male community members. A few people mentioned natural remedies (fruit, vegetables and tree leaves) as preferable to seeking care at a health facility, specifically for a disease affecting the female breast that was perceived as difficult to discuss. The data on preferring natural remedies to accessing biomedical health services, possibly from health workers of a different gender, were not elaborated on but are significant, as this might be the preferred treatment for both men and women who wish to avoid seeking health care from someone of a different gender for sensitive health problems. For women, due to their lower socio-economic status, financial control and power dynamics in the study setting, as referred to by other participants, these inequalities may be reinforced by stigmatising and sexualised diseases, which could further deter their engagement with the formal health system.

"There is a disease that affects women's breast. . . Culturally, it is treated by making the affected woman chew certain leaves of a tree, which make the disease disappear. This is also a disease that is not talked about much, but that affects our women." G12, Male Community Member, FGD

Mirroring the comments from community health workers on the difficulties of discussing culturally sensitive symptoms with community members, on the theme of "hiding" disease, several participants explained how the reluctance to reveal or disclose deters men from accessing care from the exclusively female HDA volunteers.

"I think people are afraid and they can only talk about the disease when they are at the hospital. So I'm worried that if the person affected by those diseases is a man, they will not tell the



[female] Health Development Army member and she won't be able to identify those cases easily." G12, Male Community Member, FGD

The social norm of not discussing genital symptoms with the opposite sex was repeatedly cited as a challenge to NTD service utilisation; care seeking for NTDs that affect the genitals from health workers of the opposite gender was considered problematic for all genders, though did not apply to children. Within the health facility setting, this social norm persisted, with women and men expressing a preference to interact with health workers of the same gender. These findings resonate with studies conducted in Ethiopia and other countries which found that women prefer interacting with female health professionals rather than males for "intimate care" in both community and health facility settings [54–59]. Women reported experiences of discomfort, fear and shame in situations where the patient was seen by a health worker of the opposite gender for diseases affecting the genitalia. When a disease manifested in the body part of an adult that was related to sexual activity, it was more difficult to talk about between genders due to the sexual connotations of that body part compared to diseases that do not, such as those which affect the gastrointestinal tract, for example. This resulted in stigma related to a sexual taboo.

"Most patients prefer female health workers for female diseases and male health workers for male related diseases." C02, Male Nurse, Kebele Health Post Support, Interview

"If the disease is around the genital area, they [women] are afraid of talking with men." P01, Female HEW, Interview

"The patient was female and the doctor was male. She was ashamed, tried to cover herself with a cloth, and she was talking more with me than with the doctor." B06, Female Nurse, Hospital, Interview

Health workers themselves commented that gender does not affect how they treat patients.

"Gender issues, yes—sometimes it is difficult for female patients to talk to male health workers, but the main thing is that the health worker's attitude in handling those cases." C19, Female Midwife, Kebele and Health Post Support, Interview

One recommendation that emerged from a female health worker was to establish separate consultation areas and offer patients the option to request a health professional of the same gender for diseases affecting the breasts or genitals, such as NTDs and sexually transmitted infections, with the acknowledgement that this would be challenging at the hospital level due to predominance of male doctors and the heavy workload. The majority of comments on this theme suggests that the gender norms of socially acceptable talk transcend community settings and persist in the clinic. This was expressed through repeated statements on the preference for a health worker of the same sex for both genders, from the community to the hospital level. Women's perceived shame and fear of receiving treatment from a male clinician may also deter women from seeking services from health facilities where the doctors are predominantly male. This finding is particularly important to consider when offering services at sites with all female community health workers, also when training health workers on patient rights, such as same-sex chaperones and sensitivities around managing 'taboo' diseases in health facilities in patients of the opposite gender.



Power dynamics. People generally agreed that women seek care later than men or not at all, in the context of unequal gender relations. The examples cited included the power dynamics between women and men, such as: requiring a husband's permission to disclose their disease and seek care; economic dependency on men to pay for health services; and the threat of violence if women disclose symptoms, especially for socially stigmatising conditions that affect the genitals or cause disfigurement.

"Even if I don't have evidence to say that, but females are not coming to health facility. For example, trichiasis is common in females. Females are more exposed in my experience, but they only come when the case is complicated and they may not come at all because of economic related issues." B37, Male Paediatrician, Hospital, Interview

"Males and females are not equally faced with the diseases. The women need permission to get care from her husband, because the females are influenced by males at home whether or not to disclose the disease. Fear and stigma is there. They have no money to go for treatment and come to the health centre on their own. I have seen cases who feared to come here to the health centre with signs and symptoms in the genital area, so the gender issue was big here, yes." C02, Male Nurse, Kebele Health Post Support, Interview

"I think male cases are treated earlier than females because most of the time, females tolerating even if they have a lot of diseases, especially in the rural area. This may be due to gender violence in my opinion. I have seen that of those that come to get services, almost all are male." B34, Female Health Officer, Triage, Hospital, Interview

Although the economic costs of NTDs, such as loss of earnings due to disability, have been documented [60,61] there is a lack of evidence on how this may impact different genders and factor into relational gender dynamics and health seeking behaviours.

Conclusion

Diseases may be conceptualised and gendered according to local nosologies, with important implications for health seeking practices in rural communities. Gender-related factors affect care seeking for NTDs and are sometimes described as reasons for not seeking care at all, delayed care seeking or treating NTDs with natural remedies. Moreover, health workers reiterated that women face additional challenges in seeking care due to power dynamics and access services at later stages of disease when complications are likely to be more severe. The findings echo previous studies which found that women may experience a greater NTD burden, with both social and physical consequences [9]. Also, that gender plays a role in access to financial resources, in the case of women negatively and the uptake and timeliness of care seeking [5,16], as identified in other studies [18,19, 38, 39]. Participants recommended raising community awareness about NTDs, however this approach would need to consider the gender and social norms around appropriate talk in order to be socially acceptable. Participants recommended training men to be HDA leaders to overcome this challenge, which would also support community level care for male patients with hydrocele.

Contributing to the limited literature on gender and equity in NTD research, this study provides insights on the important influence of gender in health care seeking for NTDs in local contexts. The results demonstrate that gender is significant in health seeking for NTDs, particularly those which cause genitourinary tract symptoms and are perceived as socially taboo to discuss. These findings can be useful for informing programme design towards ensuring the promotion of equitable implementation. Yet there remains a need for in-depth,



qualitative exploration of gender, how this intersects with or reinforces other social determinants, and health seeking for NTD care in different socio-economic and cultural settings. In future studies and discourse, the public health research community could progress how gender is conceptualised and discussed, for example moving away from limited descriptions of gender as binary to encompass the range of gender identities. NTD programmes which neglect gender considerations in their design and implementation risk exacerbating gender and health and health system, social and structural inequalities [44]. Those which adopt a gendered approach to NTD health service delivery can enhance acceptability and uptake, in line with the provision of universal health coverage.

Study limitations

A number of limitations of this study should be noted. Some of the data collectors had limited experience of conducting qualitative interviews and writing in English. Also, participant age was not considered in the selection criteria. No efforts were made to recruit marginalised community members such as adolescents, the elderly, or people who are stigmatised, therefore their views may not be represented by the findings. Due to limitations in terms of time and budget and as the tools could only be used meaningfully with participants who had been exposed to the small-scale intervention, it was also not possible to pre-test the topic guides. However, quality assurance was provided by a senior member of the study team throughout the field work and data were collected over the course of three iterative rounds, with improvements and adaptations made to the tools throughout the process. As the evaluation was focussed on the broader implementation of the programme, the topic guides did not include open-ended questions on the theme of equity; specifically in terms of barriers to service uptake. Lastly, the results represent the local area where the study was conducted, therefore the findings are limited in their transferability and generalisability to other contexts.

Based on our findings, we have identified the following implications for policy, practice and research:

- Consider grouping NTDs by symptoms (including similar non-NTDs where appropriate) to better accommodate how communities conceptualise the diseases.
- Include gender issues in health worker and community health worker training.
- At the national policy and local levels, promote gender balance in the selection of health workers for all NTD related training programmes, particularly HDA volunteers and Health Extension Workers.
- Design gendered health outreach activities that follow culturally appropriate norms, for
 example employing community health workers of different genders to lead sessions on sensitive health topics and diseases.
- Utilise safe, private and appropriate times and locations for knowledge sharing, such as coffee mornings, in homes, with neighbours, people who are well known, within existing social networks, to encourage people to share their own experiences with disease, if this is socially acceptable, as determined by further research.
- Mobilise male NTD champions targeting hydrocele and urinary tract symptoms in men.
- Promote the de-stigmatisation of NTDs in the community and public discourse.
- Promote timely uptake of health services among women with a focus on community dialogues about gender inequalities



- Provide NTD services from health workers of any gender when possible, based on the patient's preference.
- At the health facility level and above, provide gender appropriate areas for examination and collection of sample.
- Provide NTD services in a welcoming, safe environment at health facilities.
- Conduct a larger study which represents the country's diverse regions to inform national policy and practice on gender mainstreaming within national NTD programmes.
- Conduct further ethnographic research on gender, gender relations between community member and health workers, intersectionality and NTD services.
- Future research could explore local and contextual conceptualisations of gender. The authors encourage a shift away from narrow, binary definitions of gender to encompass all gender expressions in public health research and programming.

Supporting information

S1 Appendix. STROBE checklist. (DOC)

S2 Appendix. Intervention description. The description of the intervention is informed by the template for intervention description and replication (TIDieR) checklist and guide. (PDF)

S3 Appendix. Topic guides. (PDF)

S4 Appendix. Raw data. (PDF)

Acknowledgments

The authors gratefully acknowledge the support received from the Ethiopian Federal Ministry of Health and the Regional Health Bureau of the Southern Nations, Nationalities, and Peoples' Region. We would also like to thank our colleagues at Malaria Consortium for their guidance and support. Finally, we would like to acknowledge the work of the field researchers and the kind support of the study participants.

Author Contributions

Conceptualization: Christian Rassi, Esey Batisso, Giuseppina Ortu, Dawit Getachew.

Formal analysis: Alexandra Wharton-Smith, Christian Rassi, Rebecca King.

Funding acquisition: Christian Rassi.

Investigation: Christian Rassi, Esey Batisso, Misganu Endriyas.

Methodology: Christian Rassi, Rebecca King.

Project administration: Christian Rassi, Esey Batisso, Dawit Getachew.

Supervision: Christian Rassi, Giuseppina Ortu, Helen Counihan, Prudence Hamade.



Writing - original draft: Alexandra Wharton-Smith, Christian Rassi.

Writing – review & editing: Alexandra Wharton-Smith, Christian Rassi, Esey Batisso, Giuseppina Ortu, Rebecca King, Misganu Endriyas, Helen Counihan, Prudence Hamade, Dawit Getachew.

References

- Alemayehu B, Tomass Z, Wadilo F, Leja D, Liang S, Erko B. Epidemiology of intestinal helminthiasis among school children with emphasis on schistosoma mansoni infection in Wolaita zone, Southern Ethiopia. BMC Public Health. 2017; 17:587. https://doi.org/10.1186/s12889-017-4499-x PMID: 28633651
- Andereck JW, Kipp AM, Ondiek M, Vermund SH. Helminth prevalence among adults in rural Kenya: a stool survey for soil-transmitted helminths and schistosomiasis in Nyanza province. Trans R Soc Trop Med Hyg. 2014;(October):804–9. https://doi.org/10.1093/trstmh/tru164 PMID: 25359321
- Mewabo AP, Moyou RS, Kouemeni LE, Ngogang JY, Kaptue L, Tambo E. Assessing the prevalence of urogenital schistosomaisis and transmission risk factors amongst school-aged children around Mapé dam ecological suburbs in Malantouen district, Cameroon. Infect Dis Poverty. 2017; 6(1):40. https://doi. org/10.1186/s40249-017-0257-7 PMID: 28260525
- 4. El Katsha S, Watts S. Gender, behaviour and health: schistosomiasis transmission and control in rural Egypt. Cairo; New York: The American University in Cairo Press; 2002. 229.
- Courtright P, West SK. Contribution of sex-linked biology and gender roles to disparities with trachoma. Emerg Infect Dis. 2004; 10(11):2012–6. https://doi.org/10.3201/eid1011.040353 PMID: 15550216
- **6.** Aagaard-Hansen J, Chaignat CL. Neglected tropical diseases: equity and social determinants. In: Equity, social determinants and public health programmes. Geneva, Switzerland: WHO; 2010.
- Sow S, de Vlas SJ, Stelma F, Vereecken K, Gryseels B, Polman K. The contribution of water contact behavior to the high schistosoma mansoni infection rates observed in the Senegal River Basin. BMC Infect Dis. 2011; 11(1):198.
- Okwa OO, Tropical parasitic diseases and women. Annals of African Medicine. 2007; Vol 6, Issue 4, pp.157–63. https://doi.org/10.4103/1596-3519.55704 PMID: 18354939
- Uniting to Combat Neglected Tropical Diseases. Neglected tropical diseases: women and girls in focus. London, UK; 2016.
- Bangert M, Molyneux DH, Lindsay SW, Fitzpatrick C, Engels D. The cross-cutting contribution of the end of neglected tropical diseases to the sustainable development goals. Infect Dis Poverty. 2017; 6 (1):73. https://doi.org/10.1186/s40249-017-0288-0 PMID: 28372566
- Congdon N, West S, Vitale S, Katala S, Mmbaga BBO Exposure to children and risk of active trachoma in Tanzanian women. Am J Epidemiol. 1993; 137:366–72 https://doi.org/10.1093/oxfordjournals.aje. a116683 PMID: 8452144
- Molla YB, Le Blond JS, Wardrop N, Baxter P, Atkinson PM, Newport MJ, et al. Individual correlates of podoconiosis in areas of varying endemicity: a case-control study. PLoS Negl Trop Dis. 2013; 7(12).
- 13. Kelemework A, Tora A, Amberbir T, Agedew G, Asmamaw A, Deribe K, et al. "Why should I worry, since I have healthy feet?" A qualitative study exploring barriers to use of footwear among rural community members in northern Ethiopia. BMJ Open. 2016; 6(3):e010354. https://doi.org/10.1136/bmjopen-2015-010354 PMID: 27006343
- Aagaard-Hansen J, Mwanga JR, Bruun B. Social science perspectives on schistosomiasis control in Africa: past trends and future directions. Parasitology. 2009; 136(13):1747. https://doi.org/10.1017/ S0031182009006404 PMID: 19523256
- **15.** Hotez PJ. Empowering women and improving female reproductive health through control of neglected tropical diseases. PLoS Negl Trop Dis. 2009; 3(11).
- Dijkstra JIR, Van Brakel WH, Van Elteren M. Gender and leprosy-related stigma in endemic areas: A systematic review. Leprosy Review. 2017; 88(3): 419.
- Ayode D, Tora A, Farrell D, Tadele G, Davey G, McBride CM. Dual perspectives on stigma: reports of experienced and enacted stigma by those affected and unaffected by podoconiosis. J Public Health Res. 2016; 5(2):689. https://doi.org/10.4081/jphr.2016.689 PMID: 27747202
- Vlasso C, Weiss M, Ovuga EB, Eneanya C, Nwel PT, Babalola SS et al. Gender and the stigma of onchocercal skin disease in Africa. Soc Sci Med 2000; 50:1353–68. https://doi.org/10.1016/s0277-9536(99) 00389-5 PMID: 10741573



- Rao S, Garole V, Walawalkar S, Khot S, Karandikar N. Gender differentials in the social impact of leprosy. Lepr Rev 1996; 67:190–9. https://doi.org/10.5935/0305-7518.19960019 PMID: 8885612
- Coreil J, Mayard G, Louis Charles J, Addiss D. Filarial elephantiasis among Haitian women: social context and behavioural factors in treatment. Trop Med Int Health. 1998; 3(6): 467–473. https://doi.org/10.1046/j.1365-3156.1998.00238.x PMID: 9657509
- Pearson B, Addiss DG, Bartholomew LK, Meiger C, Pou V, Borne BV. Health seeking behaviours and self-care practices of Dominican women with lymphoedema of the leg: implications for lymphoedema management programs. Filaria J. 2006; 5: 13–21. https://doi.org/10.1186/1475-2883-5-13 PMID: 17187660
- Wijesinghe RS, Wickremasinghe AR, Sriyani E, Perera MSA. Physical disability and psychosocial impact due to chronic filarial lymphoedema in Sri Lanka. Filaria J. 2007; 6(4): 17–24.
- 23. Wynd S, Durrheim DN, Carron J, Selve B, Chaine JP, Leggat PA, Melrose W. Socio-cultural insights and lymphatic filariasis control–lessons from the Pacific. Filaria J. 2007; 6(3): 27–36.
- **24.** Alonso LM, Alvar J. Stigmatizing neglected tropical diseases: a systematic review. Soc Med. 2010; 5 (4):218–27.
- 25. Omudu E, Okafor F. Gender dimensions of knowledge, physical and psycho-social burden due to lymphatic filariasis in Benue State, Nigeria, J. Parasitol. Vect. Biol. 2011; 3: 26.
- Palmer SL, Winskell K, Patterson AE, Boubacar K, Ibrahim F, Namata I, et al. "A living death": A qualitative assessment of quality of life among women with trichiasis in rural Niger. Int Health. 2014; 6(4):291

 7. https://doi.org/10.1093/inthealth/ihu054 PMID: 25125577
- 27. Mousleya E, Deribea K, Tamiruc A, Tomczykd S, Hanlone C, Davey G. Mental distress and podoconiosis in Northern Ethiopia: a comparative cross-sectional study. Int Health. 2015; 7: 20.
- Tsegay G, Deribe K, Deyessa N, Addissie A, Davey G, Cooper M, et al. 'I should not feed such a weak woman'. Intimate partner violence among women living with podoconiosis: A qualitative study in northern Ethiopia. PLOS ONE. 2018; 13 (12): e0207571. https://doi.org/10.1371/journal.pone.0207571 PMID: 30521548
- **29.** Rilkoff H, Tukahebwa EM, Fleming FM, Leslie J, Cole DC. Exploring gender dimensions of treatment programmes for neglected tropical diseases in Uganda. PLoS Negl Trop Dis. 2013; 7(7).
- Cohn DA, Kelly MP, Bhandari K, Zoerhoff KL, Batcho WE, Drabo F, et al. Gender equity in mass drug administration for neglected tropical diseases: data from 16 countries. Int Health. 2019; https://doi.org/10.1093/inthealth/ihz012;2
- Krentel A, Gyapong M, Mallya S, Boadu NY, Amuyunzu-Nyamongo M, Stephens M, et al. Review of the factors influencing the motivation of community drug distributors towards the control and elimination of neglected tropical diseases (NTDs). PLoS Negl Trop Dis. 2017; 11(12): e0006065; 11. https://doi.org/10.1371/journal.pntd.0006065 PMID: 29211746
- Vouking MZ, Tamo VC, Tadenfok CN. Contribution and performance of female community-directed distributors in the treatment of onchocerciasis with ivermectin in Sub-Saharan Africa: a systematic review.
 Vol. 20, Pan African Medical Journal. 2015; 1–8. https://doi.org/10.11604/pamj.2015.20.1.5568 PMID: 25995798
- Brieger WR, Otusanya SA, Oke GA, Oshiname FO, Adeniyi JD. Factors associated with coverage in community directed treatment with ivermectin for onchocerciasis control in Oyo State, Nigeria. Trop Med Int Heal. 2002; 7(1):11–8.
- **34.** Jenson A, Gracewello C, Mkocha H, Roter D, Munoz B, West S. Gender and performance of community treatment assistants in Tanzania. Int J Qual Heal Care. 2014; 26(5):524–9.
- O'Connor JO, Lynch M, Vitale S, West S. Characteristics of effective Village Treatment Assistants: the Kongwa Trachoma Project. Ophthalmic Epidemiol. 1999; 6(4):257–65. https://doi.org/10.1076/opep.6.4.257.4186 PMID: 10544340
- Katabarwa MN, Habomugisha P, Ba SA. Involvement and performance of women in community directed treatment with ivermectin for onchocerciasis control in Rukungiri District, Uganda. Heal Soc Care Community. 2002; 10(5):382–93.
- 37. Parker M, Allen T. Does mass drug administration for the integrated treatment of neglected tropical diseases really work? Assessing evidence for the control of schistosomiasis and soil-transmitted helminths in Uganda. Heal Res Policy Syst. 2011; 9(3).
- Garapati P, Pal B, Siddiqui NA, Bimal S, Das P, Murti K, et al. Knowledge, stigma, health seeking behaviour and its determinants among patients with post kalaazar dermal leishmaniasis, Bihar, India. PLoS ONE. 2018; 13(9):e0203407;.5. https://doi.org/10.1371/journal.pone.0203407 PMID: 30192805
- Pepin J, Mpia B, Iloasebe M. Trypanosoma brucei gambiense African trypanosomiasis: differences between men and women in severity of disease and response to treatment. T Roy Soc Trop Med H. 2002; 96: 421–426.



- Clemmons L, Amazigo UV, Bissek AC, Noma M, Oyene U, Ekpo U, Msuya-Mpanju J, Katenga S, Seketeli A. Gender issues in the community-directed treatment with ivermectin (CDTI) of the African Programme for Onchocerciasis Control (APOC). Anna Trop Med Parasit. 2002; 96: 59–74.
- Zodpey SP, Tiwari RR, Salodkar AD. Gender differentials in the social and family life of leprosy patients. Leprosy Rev. 2000; 71: 505–510. https://doi.org/10.5935/0305-7518.20000053 PMID: 11201906
- Coreil J, Mayard G, Louis-Charles J, Addiss D. Filarial elephantiasis among Haitian women: Social context and behavioural factors in treatment. Trop Med Int Health. 1998; 3: 467–473. https://doi.org/10.1046/j.1365-3156.1998.00238.x PMID: 9657509
- 43. Manderson L, Aagaard-Hansen J, Allotey P, Gyapong M, Sommerfeld J. Social research on neglected diseases of poverty: continuing and emerging themes. PLoS Negl Trop Dis. 2009; 3(2): e332;3. https://doi.org/10.1371/journal.pntd.0000332 PMID: 19238216
- Gazzinelli A, Correa-Oliveira R, Yang GJ, Boatin BA, Kloos H. A research agenda for helminth diseases of humans: social ecology, environmental determinants, and health systems. PLoS Negl Trop Dis. 2012; 6(4): e1603. https://doi.org/10.1371/journal.pntd.0001603 PMID: 22545168
- United Nations. Transforming our world: the 2030 agenda for sustainable development. UN Geneva;
 2015.
- World Health Organization. World Health Report 2010. Health systems financing: the path to universal coverage. WHO; 2010, URL: https://www.uhc2030.org
- 47. World Health Organization. Accelerating work to overcome the global impact of neglected tropical diseases: a roadmap for implementation. WHO Geneva; 2012.
- Hoffmann TC, Glasziou PP, Boutron I, Milne R, Perera R, Moher D et al. Better reporting of interventions: template for intervention description and replication (TIDieR) checklist and guide. BMJ. 2014; 348: g1687. https://doi.org/10.1136/bmj.g1687.PMID: 24609605
- 49. Malaria Consortium. Strengthening primary healthcare for neglected tropical diseases in Ethiopia. London, UK: Malaria Consortium; 2018.
- Tong A, Sainsbury P, Craig J. Consolidated criteria for reporting qualitative research (COREQ): a 32item checklist for interviews and focus groups. Int J Qual Health Care. 2007; 19 (6) 349–357. https://doi.org/10.1093/intghc/mzm042 PMID: 17872937
- 51. Holstein J, Gubrium J. The Active Interview. USA: SAGE Publications. 1995.
- Kitzinger J. The methodology of focus groups: the importance of interaction between research participants. Sociol Health III. 2008; 16(1)103.
- 53. Glaser BG, Strauss AL. The Discovery of Grounded Theory: Strategies for Qualitative Research. New York: Aldine De Gruyter; 1967.
- 54. Mumtaz Z, Salway S, Nykiforuk C, Bhatti A, Ataullahjan A, Ayyalasomayajula B. The role of social geography on Lady Health Workers' mobility and effectiveness in Pakistan. Soc Sci Med. 2013; 91: 48–57. https://doi.org/10.1016/j.socscimed.2013.05.007 PMID: 23849238
- 55. Maes K, Closser S, Vorel E, Tesfaye Y. Using community health workers: Discipline and Hierarchy in Ethiopia's Women's Development Army. Ann Anthropol Pract. 2015; 39: 42–57.
- 56. Kok M, Kea AZ, Datiko D et al. A qualitative assessment of health extension workers' relationships with the community and health sector in Ethiopia: opportunities for enhancing maternal health performance. Hum Resour Health. 2015; 13:80. https://doi.org/10.1186/s12960-015-0077-4 PMID: 26423049
- 57. King R, Jackson R, Dietsch E, Hailemariam A. Barriers and facilitators to accessing skilled birth attendants in Afar region, Ethiopia. Midwifery. 2015; 31: 540–546 https://doi.org/10.1016/j.midw.2015.02. 004 PMID: 25745841
- Jackson R, Tesfay F, Godefay H, Gebrehiwot T. Health extension workers' and mothers' attitudes to maternal health service utilization and acceptance in Adwa Woreda, Tigray Region, Ethiopia. PLoS ONE. 2016; 11: e0150747. https://doi.org/10.1371/journal.pone.0150747 PMID: 26963507
- Jackson R, Kilsby D, Hailemariam A. Gender exploitative and gender transformative aspects of employing Health Extension Workers under Ethiopia's Health Extension Program. Trop Med & Int Health, March 2019; 24:3), pp.304–319.
- Bukachi SA, Wandibba S, Nyamongo IK. The socio-economic burden of human African trypanosomiasis and the coping strategies of households in the South Western Kenya foci. PLoS Negl Trop Dis. 2017; 11(10), e0006002. https://doi.org/10.1371/journal.pntd.0006002 PMID: 29073144
- 61. Stillwaggon E, Sawers L, Rout J, Addiss D, Fox L. Economic costs and benefits of a community-based lymphedema management program for lymphatic filariasis in Odisha State, India. Am J Trop Med Hyg. 2016; 95(4), 877–884. https://doi.org/10.4269/ajtmh.16-0286 PMID: 27573626